Author's response to reviews

Title: ZDHHC8 as a candidate gene for schizophrenia? Analysis of a putative functional intronic marker in case-control and family-based association studies

Authors:

Thomas Faul (thomas.faul@gmx.de)
Micha Gawlik (gawlik_m@klinik.uni-wuerzburg.de)
Martin Bauer (reme005@mail.uni-wuerzburg.de)
Sven Jung (sven.jung@mail.uni-wuerzburg.de)
Bruno Pfuhlmann (pfuhlmann_b@klinik.uni-wuerzburg.de)
Burkhard Jabs (burkhard.jabs@mail.uni-wuerzburg.de)
Michael Knapp (knapp@uni-bonn.de)
Gerald Stober (stoeberrg@klinik.uni-wuerzburg.de)

Version: 2 Date: 25 August 2005

Author's response to reviews:

Review 1:
Major revisions:
For the intended rapid communication we waved the two sentences which were unclear to the reviewer(1st paragraph of the discussion; incl. 3 references).
Minor revisions: we corrected the 4 suggestions

Reviewer 2 No suggestions had been made